Author’s response to reviews

Title: Comparison of modified MAZE with minimally invasive monopolar ablation and traditional bipolar radiofrequency ablation in the treatment of atrial fibrillation

Authors:

Wei Si (weisi@zju.edu.cn)
Sijia Yang (3130000773@zju.edu.cn)
Linhui Pan (lhpan@zju.edu.cn)
Chengcheng Li (chengchengli007@gmail.com)
Liang Ma (ml1402@zju.edu.cn)

Version: 2 Date: 28 Jul 2019

Author’s response to reviews:

To reviewer #1:

There are one attending doctor and 1 or 2 assistants.

Patients chose either procedures by price, severity or doctors' advise.

Surgeons perform both open and minimally invasive.

Bilateral maze procedure can only be performed in open surgery because of the limited view in minimally invasive surgery.

To reviewer #2:

English sentences have been polished.